Author's response to reviews

Title: Fidelity and moderating factors in complex interventions: A case study of a continuum of care program for frail elderly people in health and social care

Authors:

Henna Hasson (henna.hasson@ki.se)
Staffan Blomberg (staffan.blomberg@soch.lu.se)
Anna Dunér (anna.duner@socwork.gu.se)

Version: 3 Date: 1 March 2012

Author's response to reviews: see over
Manuscript 6471451675873162
Fidelity and moderating factors in complex interventions: A case study of a continuum of care program for frail elderly people in health and social care

We have accepted all the changes suggested by the copy-editor and send a clean version of the manuscript. The reference in figure legend refers to reference number 6.

Sincerely yours

Henna Hasson, PhD

Vårdal Institute

Swedish Institute for Health Sciences,

Lund University,

P.O. Box 187, SE-221 00 Lund, Sweden